Author's response to reviews

Title: Sasang Constitution Affects the Incidence of Functional Dyspepsia

Authors:

Yoon Jeong Kim (kyj86314@naver.com)
Yo Chan Ahn (ycahn@dju.kr)
Chnag Gue Son (ckson@dju.ac.kr)

Version: 2 Date: 27 February 2015

Author's response to reviews:

1. We include line and page numbers into our manuscript as you required.
2. The header (Evidenced Complementary and Alternative Medicine) is typo error. All authors re-certify that this paper has not been considered for publication elsewhere.